Author's response to reviews

Title: Complications of circumcision in male neonates, infants and children: A systematic review

Authors:

Helen A Weiss (helen.weiss@lshtm.ac.uk)
Natasha Larke (natasha.larke@lshtm.ac.uk)
Daniel Halperin (dhalperi@hsph.harvard.edu)
Inon Schenker (inon@operation-ab.org)

Version: 3 Date: 19 January 2010

Author's response to reviews:

Responses below. However, I am not sure how to refigure the tables to fit on one page of A4 and would appreciate your advice. They are core to the paper and cannot be given as simply additional tables. Clearly one option is to reduce the number of columns but all the information given is central to the interpretation of the findings.

Formatting changes to be made:

----------------------------

Major revisions (we require the author to make these changes)

Structure: this manuscript does not have the correct structure. Please reformat it to use these headings: Please insert a heading titled 'Conclusion' after the Discussion.

DONE (page 9)

Authors' contributions: Please indicate that all authors read and approved the final manuscript.

DONE (page 11)

Tables: tables 1 and 2 have been divided into parts, a and b. It is important that tables are numbered in ascending numerical order: 1, 2 and 3. Please renumber the tables and any references to them in the text. Alternatively, please merge the tables into one larger table.

DONE - Table 1 and 2 have been split and tables renumbered Tables 1-6. The text has been altered accordingly.

Tables: tables should fit into an A4 page (maximum size 176mm x 230mm),
preferably portrait.

If you cannot re-orientate the table to fit this size, please move it to the additional files. Tables in the additional files will be made available with the final manuscript upon publication but will not be included in the manuscript in situ. Please ensure that you update the text to cite additional files and include a legend in the additional files section.

I HAVE TRIED TO SHORTEN THE TABLES, BUT CANNOT MAKE TABLES 1, 3 and 6 FIT ON ONE PAGE OF A4. HOWEVER, THESE ARE CENTRAL TO THE MANUSCRIPT AND SHOULD NOT BE ADDITIONAL FILES. WOULD APPRECIATE YOUR ADVICE.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Title page: Please present the authors' names and the email addresses in separate sections on the title page.
DONE (page 1)

Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, 'The Title of My Prestigious Publication: A Scientific Study.' should be changed to 'The title of my prestigious publication: a scientific study'.
DONE: Page 1

Affiliations: Please remember to include the country in the affiliation details.
DONE: Page 1

Title page: Please remove the word count and running head from the title page.
DONE

List of abbreviations: Please rename this section 'Abbreviations'.
DONE (page 11)

Abbreviations: Please format these in to a paragraph.
DONE (page 11)

References: please provide the full citation details for references 15, 31, 32, 48 and 73. You can view the BMC reference guide at the following link http://www.biomedcentral.com/bmcgenet/ifora/#references.
15 - this is the complete reference, there is no abstract number

31 - completed

32 - completed

48 - completed

73 - completed

Tables- please remove the visible vertical lines from your tables.

DONE

Please remove the blank pages 21 and 26.

DONE

Additional file: Please remove this from the manuscript.

DONE

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

DONE

Thank you